Author's response to reviews

Title: The association between the apolipoprotein A1/ high density lipoprotein -cholesterol and diabetes in Taiwan- A cross-sectional study

Authors:

Zhi-Hong Jian (s841039@yahoo.com.tw)
Chia-Chi Lung (dinoljc@csmu.edu.tw)
Pei-Chieh Ko (ko52070@hotmail.com)
Yi-Hua Sun (iris.yihuasun@gmail.com)
Jing-Yang Huang (wchinyang@yahoo.com.tw)
Chien-Chang Ho (ccho1980@gmail.com)
Chia-Yo Ho (huc780908@hotmail.com)
Chien-Jen Chen (chencj@gate.sinica.edu.tw)
Yung-Po Liaw (Liawyp@csmu.edu.tw)

Version: 4 Date: 28 June 2013

Author's response to reviews: see over
Dear Dr. Stavros Liatis:

Enclosed is the revised manuscript entitled "The association between the apolipoprotein A1/ high density lipoprotein -cholesterol and diabetes in Taiwan — A cross-sectional study". We have adopted essentially all the suggestions made by you and the reviewer in the revision.

Responses to comments by you and the reviewer are as followings:

1. Please clarify in the "Methods"why an A1c>6 has been used as cutoff for diabetes.

We thank the valuable suggestions made by you and the reviewer. We had adopted A1c≥6.5% as cutoff for diabetes and rewritten our results. The lipid index of ApoA1/HDL remained significant with a linear response in both sexes.

Sincerely,

Yung-Po Liaw

Department of Public Health and Institute of Public Health, Chung Shan Medical University, No. 110, Sec. 1, Chien-Kuo N. Road, Taichung City 40201, Taiwan

R.O.C

Phone: +88-642-4730022 ext.11838

Fax: +88-642-3248179

Email: Liawyp@csmu.edu.tw